Author's response to reviews

Title: Spinal cord injury secondary to high-voltage electrocution: a case report.

Authors:

Harpreet K. Johl (harpreetjohl@gmail.com)
Adel Olshansky (adel.olshansky@gmail.com)
Said R. Beydoun (sbeydoun@usc.edu)
Richard A. Rison (rison@usc.edu)

Version: 3 Date: 5 March 2012

Author's response to reviews: see over
March 5th, 2012

Dear JMCR Editorial Team,

We appreciate the Deputy Editor’s comments. Below is a point-by-point response:

1. “Please include the ethnicity of the patient in the case presentation section of the manuscript.” *The ethnicity was stated in the abstract, but not the case presentation of the paper.* We now state the patient’s ethnicity within the case presentation of the manuscript as well.

2. “The question that I was not able to answer having read this manuscript was what this case report told us that we did not already know. It does seem to me that a substantial number of cases of Factor VII deficiency have been reported in the literature. So my question is how this case report significantly advance our understanding of isolated Factor VII deficiency?” *There must be a misunderstanding here, as this is not the topic of our manuscript.*

Sincerely yours,

Richard A. Rison, M.D., F.A.A.N.E.M.